Author's response to reviews

Title: A precautionary public health protection strategy for the possible risk of childhood leukaemia from residential exposure to power frequency magnetic fields

Authors:

Myron P Maslanyj (myron.maslanyj@hpa.org.uk)
Tracy Lightfoot (Tracy.Lightfoot@equ.york.ac.uk)
Joachim Schüz (joachim@cancer.dk)
Zenon Sienkiewicz (zenon.sienkiewicz@hpa.org.uk)
Alastair McKinlay (alastair.mcKinlay@hpa.org.uk)

Version: 2 Date: 1 July 2010

Author's response to reviews:

Dear Sir/Madam
There has been much concern in recent years about the possible increased risk of childhood leukemia arising from exposure to electromagnetic fields associated with the electricity supply. This paper provides an up to date summary of the status of the science and the discussions on preventive action plans.
I hope it will be attract interest in your journal.
Yours faithfully
Dr Myron P Maslanyj
EMF Dosimetry Group
Health Protection Agency
UK